Author's response to reviews

Title: MicroRNA-34a modulates genes involved in cellular motility and oxidative phosphorylation in neural precursors derived from human umbilical cord mesenchymal stem cells

Authors:

Shing-Jyh Chang (justine3@ms8.hinet.net)
Shun-Long Weng (a4467@ms7.mmh.org.tw)
Jui-Yu Hsieh (juiyu.hsieh@gmail.com)
Tao-Yeuan Wang (tywang@ms2.mmh.org.tw)
Margaret Dah-Tsyr Chang (lscmdt@life.nthu.edu.tw)
Hsei-Wei Wang (hwwang@ym.edu.tw)

Version: 8 Date: 9 September 2011

Author's response to reviews: see over
Title: “MicroRNA-34a modulates genes involved in cellular motility and oxidative phosphorylation in neural precursors derived from human umbilical cord mesenchymal stem cells”

Dear Dr. Cho,

Thank you again for your assistance on our previous revision. We now resubmit the revised manuscript. We have changed some sentences that were in red into black for publication. If there are any more questions, please do not hesitate to contact me directly.

Yours sincerely,

Hsei-Wei Wang, Ph.D.
Associate Professor
Institute of Microbiology and Immunology
National Yang Ming University
No. 155, Sec 2, Li-Nong Street
Taipei 112
Taiwan
Tel: +886-2-2826-7109
FAX: +886-2-2821-2880
Email: hwwang@ym.edu.tw